Check for updates

OPEN ACCESS

EDITED BY Sally Lau, New York University, United States

REVIEWED BY Naomi Fujioka, University of Minnesota Twin Cities, United States Nicola Petragnani, Asl Lanciano Vasto Chieti, Italy

*CORRESPONDENCE Remziye Zaim Zaim@eshpm.eur.nl; Zaim.remziye@gmail.com

SPECIALTY SECTION

This article was submitted to Thoracic Oncology, a section of the journal Frontiers in Oncology

RECEIVED 12 September 2022 ACCEPTED 06 February 2023 PUBLISHED 08 March 2023

CITATION

Zaim R, Redekop WK and Uyl-de Groot CA (2023) Incorporating risk preferences of patients in the valuation of immune checkpoint inhibitors for non-small cell lung cancer. *Front. Oncol.* 13:1027659. doi: 10.3389/fonc.2023.1027659

COPYRIGHT

© 2023 Zaim, Redekop and Uyl-de Groot. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Incorporating risk preferences of patients in the valuation of immune checkpoint inhibitors for non-small cell lung cancer

Remziye Zaim ()*, W. Ken Redekop and Carin A. Uyl-de Groot

Erasmus School of Health Policy & Management, Erasmus University, Rotterdam, Netherlands

Immunotherapy offers a distinctive mechanism of action compared to traditional treatments, arising from additional value dimensions that may not be captured in standard health technology assessments. Cancer patients may have the expectation that immunotherapy provides durable, long-term survival gains. Moreover, some patients may be willing to take a 'risk' to undergo immunotherapy to achieve better survival outcomes. We reviewed quantitative methods that explored patients' risk preferences in their non-small cell lung cancer (NSCLC) treatment choices, in PubMed (MEDLINE), from January 1, 2015, until July 1, 2022. The consideration of a value dimension ('hope') based on patients' risk-seeking preferences is specifically addressed for the valuation of immune checkpoint inhibitors in NSCLC. We reported that the quantitative methods that aim to measure patients' risk preferences or 'hope' empirically are emerging. Value assessments should not only comprise survival improvements for the mean or median patient but also consider methods that reflect durable, long-term overall survival gains for risk-seeking patients. However, the published evidence for incorporating 'hope' based on patients' stated preferences for uncertain treatment profiles is not strong, and future research could strengthen this evidence base. We encourage further research on the development and validation of quantification methods to incorporate 'hope' and risk preferences of patients treated with immunotherapy for NSCLC and beyond.

KEYWORDS

immune checkpoint inhibitor, non-small cell lung cancer, value assessment, risk preferences, hope

1 Introduction

Immunotherapy represents a significant breakthrough in the treatment of cancer. Immune checkpoint blockade is an effective therapeutic strategy that harnesses the immune system to generate an antitumor response (1, 2). Immune checkpoint inhibitors (ICIs) targeting cytotoxic T lymphocyte-associated protein 4 (CTLA-4) and programmed

cell death protein 1/programmed cell death ligand 1 (PD-1/PD-L1) have been integrated into the standard of care for patients with various cancers, including non-small cell lung cancer (NSCLC) (2). To explore the clinical efficacy and safety of these ICIs, thousands of clinical trials are underway (3). Clinical trial guidelines promote transparent and accurate reporting of patient-reported outcomes (PROs), in an effort to facilitate the interpretation and limitations of complex patient data (4, 5). Also, there is evidence that monitoring treatment side effects in real-time can improve outcomes for patients with cancer, including a potential benefit in survival rates (2, 6). However, patients may express their preferences for innovative durable therapies (e.g., immune checkpoint inhibitors) with uncertain levels of benefit, with a likelihood of a good outcome (7). Moreover, some patients (i.e., risk-seeking patients) may be willing to take additional risks (at end of life situations) to increase the probability of a survival outcome. This preference could be attributed to the evidence that although individuals may in general be risk averse, in situations where they face very poor prospects, they may become risk-seeking (8-10).

Recently, there have been theoretical research efforts to consider additional dimensions of benefit, including patients' risk preferences in their treatment choices (11). Value assessment frameworks quantifying clinical and economic outcomes of health technologies are often used to quantify the net value of NSCLC therapies (12). For example, the value frameworks used by the National Institute for Health and Care Excellence (NICE) and the Institute for Clinical and Economic Review (ICER) use average health-related quality of life (HRQoL) as a key measure of health benefit (13–15). Similarly, the value framework of the European Society for Medical Oncology enables optional weighting of treatment outcomes based on HRQoL (16). Moreover, the value framework of the American Society of Clinical Oncology (ASCO) recognizes the value of additional survival gains as part of their evaluation process (17).

Quality-adjusted life year (QALY), as a health benefit measure, is based on the assumption that marginal utility equals average utility in both quality of life and life year, and that the utility is linear, not concave, or equivalently that patients are risk-neutral (9). Introduction of risk preferences may provide a way to incorporate variability in health benefits that have not been incorporated in standard value assessments. To address the potential impact of incorporating patients' risk-seeking preferences in value assessment frameworks, we reviewed the evidence on quantitative methods and reflected upon their potential incorporation in the valuation of ICIs in NSCLC. A consideration of 'hope' as a value dimension based on patients' risk-seeking preferences is specifically addressed.

2 Identification of valuation methods for patients' risk preferences

We searched quantitative methods that explored the risk preferences of patients in their cancer treatment choices, in PubMed (MEDLINE), from January 1, 2015, until July 1, 2022. (see Supplementary Table 1) The consideration of a value dimension ('hope') based on patients' risk-seeking preferences is specifically addressed for the valuation of ICIs in NSCLC. Study findings were summarized descriptively.

3 Consideration of risk preferences in standard value assessment

We explored standard value assessment foundations and investigated methods specifically for patients' risk preferences in their treatment choices. Although the standard value assessment frameworks provide a useful starting point, in the absence of broader value considerations, some limitations may occur. These limitations could potentially lead to suboptimal resource allocation decisions, such as distorted signals to innovators, and imprecise evaluation of durable medical technologies (i.e., ICIs) (11). Although there are strengths of the standard value assessments that use the 'incremental costeffectiveness ratio' methodology, a number of limitations exist as well. In the standard value assessments, average health benefits and costs are included in the valuation of technologies. However, patient preferences and clinical practice may differ from such average outcomes. While HTA agencies use a number of criteria to make coverage decisions, institutions such as the NICE in the United Kingdom focus on cost-effectiveness and affordability as key determinants of their appraisal decisions (18). Although healthcare budgets are limited in such jurisdictions, there is a strong desire to reimburse innovative cancer therapies with some uncertainty, as demonstrated by the Cancer Drug Fund of the National Health Services in England (19). Similarly, Canada has a distinct review process for reimbursement of oncology drugs. The Canadian Agency for Drugs and Technologies in Health pan-Canadian Oncology Drug Review is responsible for the assessment of cancer treatments (20). The methods used by these agencies, however, do not explicitly capture patients' risk preferences or incorporate 'hope' for risk-seeking patients.

3.1 "Hope" for risk-seeking patients

The consideration of 'hope' as a value dimension based on patients' risk-seeking preferences is becoming evident for the valuation of therapies in NSCLC. However, we found that there are a number of definitions that may include 'hope' in the valuation of health technologies in the context of risk. Figure 1 shows multiple definitions of 'hope' identified in this context. Regardless of its nomenclature, the value of choosing among treatments with different clinical profiles, especially when some patients are willing to take the risk for a small chance of durable survival benefit or a potential cure, creates conditions for further research.

3.2 Valuation of risk preferences in patients treated with ICIs for NSCLC

To explore the risk preferences of patients for durable overall survival benefits, Shafrin et al., 2017 prospectively surveyed lung



and melanoma cancer patients and their physicians (21). For the purposes of our study, however, we solely described methods and findings pertaining to lung cancer. The authors specifically compared physicians' view of a chance of durable survival (at the tail of the survival curve), independent of average survival, to that of the patients. The results of this survey were used to determine "how patients and their physicians value therapies that offer a likelihood of durable survival outcomes" (21). "Durable survival treatments were calibrated based on survival outcomes (i.e., 66 months of follow-up) from the pivotal trials of nivolumab investigated in patients with advanced NSCLC" (18). The primary end-point was "the proportion of respondents who selected a therapy with a variable survival profile, (with some patients experiencing longterm durable survival and others experiencing shorter survival), compared to a therapy with a fixed survival duration" (21). Fixed survival was hypothetical, where "all patients were assumed to live for a specified period of time prior to their death" (21). Parameter estimation by sequential testing ("PEST") was applied to "calculate to estimate the duration of survival that would make patients or physicians indifferent between fixed survival and therapy with durable survival." (21) "PEST is an adaptive elicitation technique that determines the stimulus value for each new question using responses to the previous question." (21, 22) In the study, patients, and physicians continued to receive questions until an indifference point was reached, or until 10 questions were answered (21).

Overall, the analysis comprised 84 lung cancer patients and 96 physicians (21). There were two primary endpoints: "1) whether the respondent preferred a durable survival therapy compared with a fixed survival therapy"; and "2) the indifference point in terms of survival between a durable survival therapy and a fixed survival therapy." (21) For lung cancer, "65.5% of patients preferred the therapy with a variable survival profile, compared with 40.8% of physicians (Δ =24.7%; P 0.001)." (21) "Patients' indifference point indicated that therapies with a variable survival profile were preferred unless the treatment with fixed survival had 11.6 months longer mean survival" (21). "Physicians were prescribing treatments with a fixed survival if the treatment had 1.0 months shorter survival compared to the uncertain survival profile" (21). Based on risk preference distributions of the Kaplan-Meier curve estimations, Shafrin et al. assumed a constant relative risk aversion utility function (23). To compare the indifference point with the certainty equivalent, t-tests were performed. Patients' indifference point among lung cancer therapies with durable and fixed survival was "41.6 months (i.e., 11.6 months greater than the average survival at 30 months)" (21). In contrast, the physicians' indifference point was 29 months (21). The overall "indifference point was 12.6 months greater (P < 0.001) for patients compared to the physicians." (21) Applying a constant relative risk aversion (RRA) utility function, the authors estimated that "patients are risk-seeking (RRA = +0.39 for NSCLC; P < 0.001), and physicians are risk neutral for lung cancer treatments (RRA = -0.03; P = 0.523)." (21) "The patient's utility function was u(x) = $x^{1.39}$ and the physician's utility function was u(x) = $x^{0.97}$ " (21).

Shafrin et al. showed that "lung cancer patients were willing to give up 38.7% of an average survival for a likelihood of durable survival" (21). The patient preferences reported in this study are consistent with the prospect theory, which predicts that people may be risk-seeking in circumstances starting below their reference point (24, 25). However, it should be noted that discordant preferences do not necessarily mean that "physicians override their patients' desires" (26). Some patients may prefer that their physicians make treatment decisions on their behalf (27). Nonetheless, physicians' decisions may not always be aligned with that of patients' interests (28). Therefore, it is important to conduct additional research to understand 'when, why, and how' physicians may implicitly or explicitly substitute their perspectives on patients' behalf.

Shafrin et al., 2018 examined whether incorporating additional value considerations was influential on the cost-effectiveness estimates (26). Previous research suggested that broader societal benefits could be considered in value assessments (29, 30). For example, assessments could potentially include patients' treatment preferences that offer durable survival benefits, instead of average outcomes (21, 31). Building upon a previously published cost-effectiveness analysis (32), Shafrin et al. studied patients in Canada with advanced squamous NSCLC treated with second-line nivolumab. The authors used the net monetary benefit framework (33), to calculate cost-effectiveness estimates (34). The authors conducted their analyses from three perspectives; namely the "traditional payer", "traditional societal" and "broad societal" perspectives (34). The traditional payer perspective was built based on a model developed in Canada by the Goeree et al. (32) Goeree et al. extrapolated progression-free survival (PFS) and overall survival (OS) Kaplan Meier curves from the CheckMate 017 Phase 3 clinical trial (35). Goeree et al. estimated the proportion of patients in "progression-free, progressed disease and death" health states, for a time horizon of 10 years (32). The CheckMate 017 trial investigated the clinical and safety outcomes of "nivolumab (3 mg/kg every two weeks) compared to docetaxel (75 mg/m² every three weeks) for previously treated patients with advanced squamous NSCLC" (35). The results of this trial showed "significant improvements in median OS and PFS for nivolumab patients compared to docetaxel patients (OS: 9.2 vs. 6.0 months, HR: 0.59; PFS: 3.5 months vs. 2.8 months, HR: 0.62)." (35) Goeree et al. reported that based on the CheckMate 017 clinical trial results, the incremental cost-effectiveness ratio (ICER) was " Canadian dollars (CA)\$151,560 per QALY gained" (32).

To explore a broader societal perspective, Shafrin et al. quantified 'hope' in addition to caregiver burden, insurance value, and option value of the nivolumab treatment in patients with

advanced NSCLC. For the purposes of our review, however, we focused our descriptive summary on the valuation of 'hope' or patients' risk preferences. The Checkmate 017 clinical trial showed that the "two-year overall survival for nivolumab was 24%, and reduced to 16% at five years" (35). To quantify 'hope' (31) the authors first measured the difference in the expected survival between nivolumab and docetaxel using the Kaplan-Meier curve obtained from Goeree et al. (32) Subsequently, they estimated the "certainty equivalent between the two treatments using a utility function based on the previously reported risk aversion estimates for NSCLC patients" (21). The authors assumed that the difference between the expected survival difference and this estimated certainty equivalent provides a valuation method to quantify 'hope'. Using this method, Shafrin et al. estimated an "additional QALY gain of 0.039, beyond the baseline estimate of 0.66 QALYs" (32), at an additional cost of CA\$ 5,850 (26). The ICERs for nivolumab compared to docetaxel were "\$151,560, \$141,344, and CA\$80,645 per QALY gained from the traditional payer, traditional societal, and broader societal perspectives, respectively" (26).

Similarly, the researchers at the Innovation and Value Initiative (IVI) in the US developed a patient-centered value assessment, which incorporated 'hope' into a cost-effectiveness model for advanced NSCLC (36). In this model, treatment regimens for the study population comprised ICIs for the progressed disease after second-line therapy. From a clinical perspective, this model requires an update based on the modifications in the latest NSCLC guidelines. From an economic perspective, in the IVI's four-state model, patients were assumed to begin first-line (1L) treatment in stable disease (S1) and can either experience disease progression and consequently transition to second-line (2L) treatment (P1/S2), or death (D) (36). With 2L treatment, patients can either experience disease progression (P2) or death (D). At P2, patients begin 2L+ treatment (including ICIs) and remain in this state until death (36). IVI's NSCLC model specifically focused on the quantification of 'hope' to address patients' risk attitudes when "treatments with equivalent expected health benefits differ in their overall benefit distributions" (36). This is assumed to be true when a benefit distribution has a longer-term survival for some patients. The authors of this study quantified 'hope' as the "difference between expected incremental QALYs (based on the mean benefit) and the certainty equivalent" (36). The certainty equivalent was then defined as "the number of QALYs that a patient would need to obtain to be indifferent between the comparator and comparative treatment strategy, with alternative distribution of survival outcomes" (21, 36). This definition was again based on the Shafrin et al. study which showed that "patients place a high value on treatments with a higher probability of durable overall survival benefits" (21).

Although these NSCLC case studies illustrate potential methods to quantify patients' risk preferences or 'hope' when treated with an ICI, there are a number of methodological and practical issues that require careful consideration. Standard value assessments including cost-effectiveness assume risk neutrality (37). Ignoring risk preferences may underestimate or overestimate the value of interventions for different indications (9, 10). Consideration of two competing interventions with the same average survival, one with greater uncertainty, would be deemed comparable in a conventional or standard cost-effectiveness assessment. For some patients, however, this uncertainty may indicate that they have a treatment preference. As quantified by Shafrin et al. (21), the certainty equivalent may comprise health benefits (i.e., QALYs) that a patient may be indifferent among ICIs or the alternative treatment strategy. Estimations of 'hope' may indicate that the distribution of health benefits should not only be characterized by their variance but also by their skewness. For risk-seeking patients, including those with NSCLC, some may prefer the treatment option with durable survival, at the expense of a higher risk of dying earlier (31). In the Shafrin et al. case study (26), there were data limitations specific to nivolumab or NSCLC to measure broader societal benefits. Even if such data were available, it may not be possible that the studies would cover all indications at the time of value assessments (26). Moreover, after clinical trial results are published, there is significant uncertainty related to the long-term benefits of ICIs, owing to the relatively short follow-up period of most immunotherapy clinical trials. Therefore, payers and the HTA agencies may under or overestimate treatment value, extrapolate value estimates from different studies, indications, and conditions, or consider alternative assumptions. Nonetheless, these case studies illustrate that quantification of patients' risk preferences, referred to as 'hope' is a research area that requires further investigation and validation in NSCLC patients, specifically for those treated with an ICI.

3.3 Valuation of patients' risk preferences beyond immunotherapy in NSCLC

Studies included in our review also highlighted potential ways to characterize patients' risk preferences ("hope") beyond NSCLC. Valuation of patients' risk preferences or 'hope' in the assessments of health technologies has yet to gain traction, in part due to reliance on established practices, and the lack of motivations to improve existing frameworks (38). An alternative viewpoint is that the use of standard value assessment methods is sufficient, because "cost-effectiveness estimates are only an input to, and not a substitute for, a deliberative decision-making process that allows for additional elements of value to be contextualized into the process without the need for formal quantification" (39). Regardless of the viewpoints, patients' risk preferences, their relative importance for some indications, valuation methods, and potential quantification methods remain areas for further research. In this section, we highlight key studies that provide distinct methods to help quantify 'hope' and incorporate patients' risk preferences beyond the treatment of NSCLC.

Lakdawalla et al., estimated patients' risk preferences in melanoma, breast cancer, and other solid tumor patients (31). The authors recommended incorporating 'hope" into the valuation of end-of-life treatments or considering a higher cost-effectiveness threshold for treatments at the end of life. The authors surveyed patients' preferences on two treatment choices; "one offering a modest length of survival, and the other offering a 50% chance of a substantially longer survival, but also a 50% chance of no additional survival" (31).

If patients care about long-term survival prospects, not just average survival, this study suggests the need to incorporate the valuation of risk preferences as a unique consideration in the HTA. Arguably, a two-step approach could be considered (40). First-step may comprise a standard cost-effectiveness assessment based on average clinical survival and other health benefits (i.e., QALY) estimates. Second-step may include other value considerations, both qualitatively and quantitatively (40). However, this can be achieved based on the assumption that patients acknowledge and can act on their own (risk) preferences. All in all, Lakdawalla et al. highlighted the view that value should incorporate the "perspective of the patient", and value assessment frameworks should take this perspective into account.

Lakdawalla and Phelps (10), examined additional value elements, including 'hope', using a "generalized risk-adjusted cost-effectiveness (GRACE) model, which assumes that patients are utility maximizers in their choices about their treatment choices" (9, 10, 41). Building on the study by Garber and Phelps (37) and Lakdawalla et al. (41), this study authors incorporated patients' risk preferences in treatment choices. Patients with "severe impairments and prospects of continued poor health, or those facing shorter life expectancy, were shown to have a higher willingness to pay for durable health gains" (41). Some patients may be willing to take the risk for a treatment that has a likelihood of 'cure' or 'hope' (31). The GRACE model suggested that "patients are not indifferent to the length of life or quality of life, which could indicate that marginal utility does not equal average utility" (41). The study authors indicated that "willingness to pay per QALY thresholds may need adjustments for value assessments" (41). However, it should be noted that this finding may have important implications on the 'incentives' for future innovation developers and investments (38).

Another method to quantify 'hope' was presented by using a discrete choice experiment (DCE) (42). Using a DCE for 200 patients with cancer or a history of cancer, Reed et al. (42), reported that "patients valued treatments with 5% and 10% chances of 10-year survival, independent of expected survival, although the findings did not hold in all scenarios" (42). First a pilot DCE was designed, and "participants were asked to assume that they had recently been diagnosed with cancer that had begun to metastasize" (42). Participants then had to consider choices "when expected survival was three years, with a given chance of 10-year survival, or a case with certain 3-year survival outcome (i.e., 10-year survival was zero)" (42). After piloting the DCE using general participants, cancer patients were asked to complete the online DCE survey. The study authors found that the "estimated value of 'hope' for a 5% chance of survival was on average about \$6,000, and 10% for \$12,500" (42). "With a life expectancy of 5 years, when a 20% chance of 10-year survival corresponds to 80% of an average 3.8 years, participants' choices were consistent with expectations according to utility theory and risk neutrality" (42). However, when the choices had a "scenario with a life expectancy of 2 years, where a 20% chance of 10-year survival implies an 80% chance of a 1-month survival, patients rarely chose this option" (42). The study authors highlighted that there was heterogeneity in patients' preferences across attributes. For example; "a latent class analysis, designed to identify groups with similar preferences, found four distinct groups of participants differing in terms of their sensitivity to costs and preferences for treatments enabling durable survival" (42). All in all, Reed et al. highlighted that the quantification of 'hope' is important, however, there are uncertainties about how much 'hope' may be worth and how to quantify heterogeneous preferences. Reed et al. concluded that researchers and policymakers should assess heterogeneous patterns of risk preferences and carefully consider them for resource allocation and reimbursement decisions (42).

All in all, these studies highlighted in Section 3.3 do not specifically focus on immunotherapy or NSCLC. Given the unique characteristics of immunotherapy, such as the "tail of the curve survival potential", future studies that will present methods to quantify "hope" specifically for the valuation of ICIs are encouraged.

4 Discussion

Immunotherapy offers a distinctive mechanism of action compared to traditional treatments, arising from additional value dimensions that may not be captured in standard HTA methods. In this study, we focused specifically on the patients' risk-seeking preferences because some cancer patients may prefer treatments that have a likelihood of durable survival. Based on the available evidence, our review revealed that economic value assessment methods should not only be based on survival improvement for the mean or median patient, but also on the quantification of risk preferences for durable overall survival gains. Although ICIs have distinctive characteristics that may increase the relevance of considerations of additional value dimensions, a major issue that should be considered against the inclusion of 'hope' relates to equity concerns. Higher spending on certain ICIs or other durable medical technologies that get extra importance based on 'hope' or patients' stated preferences may have consequences (i.e., opportunity costs) inside and/or outside the health systems.

Our study showed that the quantitative methods that aim to measure 'hope' empirically are emerging. There are viewpoints that support and refute the inclusion of 'hope' alongside standard measures of health gain (7). The evidence on the quantification methods for 'hope' based on patients' stated preferences for uncertain treatments is not strong and future research could strengthen this evidence base. One complexity is that the value in having access to therapies with different clinical profiles can diverge across patients, and across indications. Therefore, any attempt to add an empirical weighting for 'hope' at the population level may be premature.

In the United Kingdom, NICE includes additional criteria in the assessment of health technologies, in the end-of-life context as a modifier when considering its cost-effectiveness threshold (43). The intention is not to raise the threshold per se, but to give greater weight to QALYs achieved at the end-of-life (under certain circumstances), by focusing on the expected gain, rather than any element of 'hope'. The International Society for Health Economics and Outcomes Research (ISPOR) Task Force researchers, who studied the US Value Assessment Frameworks, identified eight elements of value, and suggested considering additional value

dimensions, including 'hope' (11). This ISPOR Task Force report recognizes, however, as did the recommendations of the Second Panel on Cost Effectiveness, that these additional value dimensions are subject to further research, and the methods for empirically integrating them into value frameworks are not yet validated (44, 45). There are also intrinsic equity concerns about "incorporating additional dimensions of value, without considering the opportunity costs and potential health losses that might be foregone". Nonetheless, in its value assessment framework covering the years 2020 to 2023, ICER stated that it would consider broader value elements such as option value, the value of hope, and scientific spillovers in the 'other benefits or disadvantages' and 'contextual considerations' sections of ICER evidence reports (14). However, such considerations do not influence costeffectiveness results or value-based prices.

Others have stated that "value elements, such as hope and fear, could legitimately affect individual decision-making but are fraught with difficulties in measurement as they relate to subjective experience and could be manipulated by the context" (39). Arguably, standard value assessments using the QALY also face measurement challenges, because the utility measurement can be a reflection of subjective experiences (38). Value elements such as 'hope' has the potential to be measured quantitatively using skewness in the outcome distribution (38). Improved measurement methods of value elements might allow modifiers to be applied quantitatively in the future, although challenges remain (38).

All in all, our review suggests the need for further research to develop and validate reliable methods for the quantification of 'hope' and valuation of risk preferences for patients treated with immunotherapy in NSCLC. The proliferation of empirical studies is promising, however, additional methods of development efforts are needed. If 'hope' were to be incorporated into the valuation of ICIs, the HTA evaluation would then need to depart from the traditional focus on average outcomes and include the notion that (some)

References

1. Topalian SL, Drake CG, Pardoll DM. Immune checkpoint blockade: A common denominator approach to cancer therapy. *Cancer Cell* (2015) 27:450–61. doi: 10.1016/j.ccell.2015.03.001

2. Zaim R, Redekop K, Uyl-de Groot CA. Analysis of patient reported outcomes included in the registrational clinical trials of nivolumab for advanced non-small cell lung cancer. *Transl Oncol* (2022) 20:101418. doi: 10.1016/j.tranon.2022.101418

3. Vaddepally RK, Kharel P, Pandey R, Garje R, Chandra AB. Review of indications of FDA-approved immune checkpoint inhibitors per NCCN guidelines with the level of evidence. *Cancers (Basel)* (2020) 12:E738. doi: 10.3390/cancers12030738

4. Calvert M, Blazeby J, Altman DG, Revicki DA, Moher D, Brundage MD, for the CONSORT PRO Group. Reporting of patient-reported outcomes in randomized trials: The CONSORT PRO extension. *JAMA* (2013) 309:814–22. doi: 10.1001/jama.2013.879

5. Brundage M, Blazeby J, Revicki D, Bass B, de Vet H, Duffy H, et al. Patientreported outcomes in randomized clinical trials: Development of ISOQOL reporting standards. *Qual Life Res* (2013) 22:1161–75. doi: 10.1007/s11136-012-0252-1

6. Basch E, Deal AM, Dueck AC, Scher HI, Kris MG, Hudis C, et al. Overall survival results of a trial assessing patient-reported outcomes for symptom monitoring during routine cancer treatment. *JAMA* (2017) 318:197. doi: 10.1001/jama.2017.7156

7. Peasgood T, Mukuria C, Rowen D, Tsuchiya A, Wailoo A. Should we consider including a value for 'Hope' as an additional benefit within health technology assessment? *Value Health* (2022) \$1098-3015(22):00148-6. doi: 10.1016/j.jval.2022.03.006

patients may care about the distribution of durable benefits, not just the average.

Author contributions

RZ: Conceptualization, data curation, writing – original draft, writing – review & editing. KR: Writing – review & editing. CU Writing – review & editing. All authors contributed to the article and approved the submitted version.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fonc.2023.1027659/ full#supplementary-material

8. Tversky A, Kahneman D. The framing of decisions and the psychology of choice. *Science* (1981) 211:453–8. doi: 10.1126/science.7455683

9. Lakdawalla DN, Phelps CE. Health technology assessment with risk aversion in health. J Health Economics (2020) 72:102346. doi: 10.1016/j.jhealeco.2020.102346

10. Lakdawalla DN, Phelps CE. Health technology assessment with diminishing returns to health: The generalized risk-adjusted cost-effectiveness (GRACE) approach. *Value Health* (2021) 24:244–9. doi: 10.1016/j.jval.2020.10.003

11. Lakdawalla DN, Doshi JA, Garrison LP, Phelps CE, Basu A, Danzon PM. Defining elements of value in health care-a health economics approach: An ISPOR special task force report [3]. *Value Health* (2018) 21:131-9. doi: 10.1016/j.jval.2017.12.007

12. Zaim R, Redekop K, Uyl-de Groot CA. Immune checkpoint inhibitors for the treatment of non-small cell lung cancer: A comparison of the regulatory approvals in Europe and the united states. *J Cancer Policy* (2022) 33:100346. doi: 10.1016/j.jcpo.2022.100346

13. National institute for health and care excellence. health technology evaluations: the manual. process and methods (2022). Available at: https://www.nice.org.uk/process/pmg36/resources/nice-health-technology-evaluations-the-manual-pdf-72286779244741 (Accessed 6 Jan 2022).

14. Institute for clinical and economic review. 2020-2023 value assessment framework (2020). Available at: https://icer.org/wp-content/uploads/2022/01/ICER_2020_2023_ VAF_120821.pdf (Accessed 4 Jan 2022). 15. Chapman R, Kumar V, Samur S, Zaim R, Segel C, Pearson SD. Institute for clinical and economic review. value assessment methods and pricing recommendations for potential cures: A technical brief (2019). Available at: https://icer.org/wp-content/uploads/2020/10/Valuing-a-Cure-Technical-Brief.pdf.

16. Cherny NI, Sullivan R, Dafni U, Kerst JM, Sobrero A, Zielinski C, et al. A standardised, generic, validated approach to stratify the magnitude of clinical benefit that can be anticipated from anti-cancer therapies: The European society for medical oncology magnitude of clinical benefit scale (ESMO-MCBS). *Ann Oncol* (2015) 26:1547–73. doi: 10.1093/annonc/mdv249

17. Schnipper LE, Davidson NE, Wollins DS, Blayney DW, Dicker AP, Ganz PA, et al. Updating the American society of clinical oncology value framework: Revisions and reflections in response to comments received. *J Clin Oncol* (2016) 34:2925–34. doi: 10.1200/JCO.2016.68.2518

18. Gettinger SN, Horn L, Gandhi L, Spigel DR, Antonia SJ, Rizvi NA, et al. Overall survival and long-term safety of nivolumab (Anti-programmed death 1 antibody, BMS-936558, ONO-4538) in patients with previously treated advanced non-Small-Cell lung cancer. *J Clin Oncol* (2015) 33:2004–12. doi: 10.1200/JCO.2014.58.3708

19. NHS England Cancer drug funds. Available at: https://www.england.nhs.uk/ cancer/cdf/.

20. Canadian Agency for drugs and technologies in health, pan-Canadian oncology drug review (pCODR). Available at: https://www.cadth.ca/pcodr-expert-review-committee-perc (Accessed 1 Jul 2022).

21. Shafrin J, Schwartz TT, Okoro T, Romley JA. Patient versus physician valuation of durable survival gains: Implications for value framework assessments. *Value Health* (2017) 20:217–23. doi: 10.1016/j.jval.2016.11.028

22. Bleichrodt H, Doctor J, Stolk E. A nonparametric elicitation of the equityefficiency trade-off in cost-utility analysis. *J Health Economics* (2005) 24:655–78. doi: 10.1016/j.jhealeco.2004.10.001

23. Pratt JW. Risk aversion in the small and in the Large. *Econometrica* (1964) 32:122. doi: 10.2307/1913738

24. Kahneman D, Tversky A. Prospect theory: An analysis of decision under risk. *Econometrica* (1979) 47:263. doi: 10.2307/1914185

25. Tversky A, Kahneman D. Advances in prospect theory: Cumulative representation of uncertainty. J Risk Uncertainty (1992) 5:297–323. doi: 10.1007/BF00122574

26. Shafrin J, Skornicki M, Brauer M, Villeneuve J, Lees M, Hertel N, et al. An exploratory case study of the impact of expanding cost-effectiveness analysis for second-line nivolumab for patients with squamous non-small cell lung cancer in Canada: Does it make a difference? *Health Policy* (2018) 122:607–13. doi: 10.1016/j.healthpol.2018.04.008

27. Sutherland HJ, Llewellyn-Thomas HA, Lockwood GA, Lockwood GA, Tritchler DL, Till JE. Cancer patients: their desire for information and participation in treatment decisions. *J R Soc Med* (1989) 82:260–3. doi: 10.1177/014107688908200506

28. McGuire TG. Chapter 9 physician agency. In: *Handbook of health economics*. Boston University, Boston, Massachusetts: Elsevier (2000). p. 461–536. doi: 10.1016/S1574-0064(00)80168-7

29. Devlin NJ, Lorgelly PK. QALYs as a measure of value in cancer. J Cancer Policy (2017) 11:19–25. doi: 10.1016/j.jcpo.2016.09.005

30. Garrison LP, Kamal-Bahl S, Towse A. Toward a broader concept of value: Identifying and defining elements for an expanded cost-effectiveness analysis. *Value Health* (2017) 20:213–6. doi: 10.1016/j.jval.2016.12.005

31. Lakdawalla DN, Romley JA, Sanchez Y, Maclean JR, Penrod JR, Philipson T. How cancer patients value hope and the implications for cost-effectiveness assessments of high-cost cancer therapies. *Health Aff (Millwood)* (2012) 31:676–82. doi: 10.1377/ hlthaff.2011.1300

32. Goeree R, Villeneuve J, Goeree J, Penrod JR, Orsini L, Monfared AAT. Economic evaluation of nivolumab for the treatment of second-line advanced squamous NSCLC in Canada: A comparison of modeling approaches to estimate and extrapolate survival outcomes. *J Med Econ* (2016) 19:630–44. doi: 10.3111/13696998.2016.1151432

33. O'Brien BJ, Briggs AH. Analysis of uncertainty in health care cost-effectiveness studies: an introduction to statistical issues and methods. *Stat Methods Med Res* (2002) 11:455–68. doi: 10.1191/0962280202sm304ra

34. Neumann P, Sanders G, Russell L, Sanders GD, Siegel JE. Cost-effectiveness in health and medicine (2nd ed.) (2016). New York: Oxford University Press 2017.

35. Brahmer J, Reckamp KL, Baas P, Crinò L, Eberhardt WEE, Poddubskaya E, et al. Nivolumab versus docetaxel in advanced squamous-cell non–Small-Cell lung cancer. N Engl J Med (2015) 373:123–35. doi: 10.1056/NEJMoa1504627

36. Innovation and value initiative. non-small cell lung cancer value model. part of the open-source value project. Available at: https://www.thevalueinitiative.org/ivi-nsclc-value-model/ (Accessed 21 Jan 2022).

37. Garber AM, Phelps CE. Economic foundations of cost-effectiveness analysis. J Health Econ (1997) 16:1–31. doi: 10.1016/s0167-6296(96)00506-1

38. Neumann PJ, Garrison LP, Willke RJ. The history and future of the 'ISPOR value flower': Addressing limitations of conventional cost-effectiveness analysis. *Value Health* (2022) 25:558–65. doi: 10.1016/j.jval.2022.01.010

39. Goring S, Garrison L, Jansen J, Yin W. Novel elements of the value flower: fake or truly novel? Available at: https://www.ispor.org/publications/journals/value-outcomes-spotlight/vos-archives/issue/view/navigating-the-changing-heor-publishing-landscape/novel-elements-of-the-value-flower-fake-or-truly-novel (Accessed 1 Jun 2022).

40. Goldman D, Lakdawalla D, Philipson TJ, Yin W. Valuing health technologies at NICE: recommendations for improved incorporation of treatment value in HTA. *Health Econ* (2010) 19:1109–16. doi: 10.1002/hec.1654

41. Lakdawalla DN, Phelps CE. A guide to extending and implementing generalized risk-adjusted cost-effectiveness (GRACE). *Eur J Health Econ Published Online First* (2021) 23:433–51. doi: 10.1007/s10198-021-01367-0

42. Reed SD, Yang J-C, Gonzalez JM, Johnson FR. Quantifying value of hope. *Value Health* (2021) 24:1511–9. doi: 10.1016/j.jval.2021.04.1284

43. National institute for health and care excellence. guidance. Available at: https:// www.nice.org.uk/about (Accessed 22 Jan 2022).

44. Sanders GD, Neumann PJ, Basu A, Brock DW, Feeny D, Krahn M, et al. Recommendations for conduct, methodological practices, and reporting of cost-effectiveness analyses: Second panel on cost-effectiveness in health and medicine. *JAMA* (2016) 316:1093–103. doi: 10.1001/jama.2016.12195

45. Carias C, Chesson HW, Grosse SD, Li R, Meltzer MI, Miller GF, et al. Recommendations of the second panel on cost effectiveness in health and medicine: A reference, not a rule book. *Am J Prev Med* (2018) 54:600–2. doi: 10.1016/ j.amepre.2017.11.013